Conc.: Submission of the paper: Prevention and pregnancy care in undocumented migrants by
Hans Wolff, Manuella Epiney, Ana P. Lourenco, Michael C. Costanza, Jacqueline Delieutraz-Marchand, Nicole Andreoli, Jean-Bernard Dubuisson, Jean-Michel Gaspoz, Olivier Irion

Dear editor-in-chief,

We are pleased to submit the above-referenced manuscript for publication in BMC Public Health.

Female undocumented migration is a growing problem worldwide. This contrasts with the scarcity of scientific knowledge about this hard-to-reach population. This study is, to our knowledge, the first comparative description of detailed information about contraception, health behaviour, violence exposure and birth outcomes, as well as maternal and neonatal health.

The findings are of interest for a broad public including public health specialists, gynaecologists, primary care and hospital-based physicians as well as politicians.

The manuscript comprises original, unpublished material which is not under consideration for publication elsewhere. All the authors have taken a significant and active part in the preparation of this article and have read and approved the final version.

Sincerely yours,